Author's response to reviews

Title: Lack of association between serological evidence of past Coxiella burnetii infection and incident ischaemic heart disease: nested case-control study

Authors:

Conall McCaughey (conall.mccaughey@bll.n-i.nhs.uk)
Liam J Murray (l.murray@qub.ac.uk)
James P McKenna (james.mckenna@bll.n-i.nhs.uk)
Peter V Coyle (peter.coyle@bll.n-i.nhs.uk)
Hugh J O'Neill (hugh.oneill@bll.n-i.nhs.uk)
Dorothy E Wyatt (dorothy.wyatt@bll.n-i.nhs.uk)
Jayne V Woodside (j.woodside@qub.ac.uk)
John WG Yarnell (j.yarnell@qub.ac.uk)
Pierre Ducimetiere (ducimetiere@vif.inserm.fr)
Annie Bingham (bingham@vif.inserm.fr)
Philippe Amouyel (philippe.amouyel@pasteur-lille.fr)
Michele Montaye (michele.montaye@pasteur-lille.fr)
Dominique Arveiler (monica@medecine.u-strasbg.fr)
Bernadette Haas (bernadette.haas@medecine.u-strasbg.fr)
Jean Ferrieres (ferriere@cict.fr)
Jean-Bernard Ruidavets (ruidavet@cict.fr)

Version: 3 Date: 19 July 2005

Author's response to reviews:

Unnecessary capital letters have been deleted from the section and subsection headings and funding details have been moved to the acknowledgements section as requested.